Author's response to reviews

Title: ZFP36 stabilizes RIP1 via degradation of XIAP and cIAP2 thereby promoting Ripoptosome assembly

Authors:

Tommaso Selmi (tselmi@gmail.com)
Claudia Alecci (claudia.alecci@gmail.com)
Miriam dell'Aquila (miriam.dellaquila@gmail.com)
Lucia Montorsi (lucia.montorsi@unimore.it)
Andrea Martello (martelloandrea@gmail.com)
Filippo Guizzetti (72295@studenti.unimore.it)
Nicola Volpi (nicola.volpi@unimore.it)
Sandra Parenti (sandra.parenti@unimore.it)
Sergio Ferrari (sergio.ferrari@unimore.it)
Paolo Salomoni (p.salomoni@ucl.ac.uk)
Alexis Grande (alexis.grande@unimore.it)
Tommaso Zanocco-Marani (zanocco@unimore.it)

Version: 5 Date: 2 April 2015

Author’s response to reviews:

Dear Editor, we checked the manuscript in order to correct English grammar mistakes and changed figures following reviewers' requests. Hoping that the paper is suitable for publication we remain.

Best regards,
Tommaso Zanocco Marani